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Notes for the February 19, 2014, CERES  

Processing Oversight Board Meeting  

 

Meeting began at 1:00 p.m. in B1268C R2316. 

Meeting Notes  

There were no comments or questions on the notes from the last meeting.   

Announcements 

The 1,000
th

 CERES SCCR was recently submitted! 

    

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting) 

PR_Database 962 – Updated to document the deferral of several items to SCCR 999. 

PR_Database 999 – Updates to the PR Tool Web Interface to address JIRA tickets 115, 116, and 

129. 

CATALYST 1000 – Delivery of the script to monitor the health and status of the CATALYST 

server. 

Inversion 1001 – Delivery of new PGE CER4.5-6.0P6.  This will be the pre-processor to 

generate Sea Ice Brightness Index for use in the Edition4 ADMs. 

 

There were no issues identified, so these SCCRs will be approved. 

 

CATALYST Status 

Actively working on three of the four RFAs.  The fourth one involves the ValRx runs after 

CATALYST is promoted. 

   

SIT is okay with the output of the recovery utility. 

   

The data required for the 8 PGEs ready to run via CATALYST are being restored to the DPO. 

 

Production Processing Status 

Tonya is working with Tom on the SYN-1deg issue.  Hoping to resolve this soon and continue 

processing. 

   

System Status   

DPO restoration effort – Projecting to finish the Category 2 list by Friday, February 21.  It’s still 

important for the DMT to check the DPO to ensure all of the requested files have been 

restored. 

 

Magneto status – There was a power issue last week and a lot of the nodes have not been 

restored and may not be.  At least half of the nodes are off line. 

 

Status of Open SCCRs 

962 – Will be closed today.  

989 – Carla is ready to send out the release notes to Maria. 

937 – Addressing the RFAs. 

996 – Will be ready to scan the DPO once the required data have been restored. 

997 – Ready to deliver. 
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1000 – Should finish the CATALYST monitoring script this week 

990 – In CM testing.  Will be released in conjunction with Perl_Lib 992 and Inversion 973.  

992 – Waiting for resolution of RT ticket 31211, which is the SGE memory consumption issue, 

before this can be delivered. 

973 – Will be compiled and tested after Perl_Lib 992 is delivered and installed. 

979 – Files required for testing have been restored to the DPO, and Angel is running the test 

cases from the Test Plan. 

919 – Files needed for testing have been restored to the DPO.  Hope to be finished testing within 

a week. 

960 –Software is converging to a solution.  Waiting for the Land PEATE to produce new VIIRS 

data (Collection 1.1).  Planning to deliver a week or two after the data files are received. 

959 – Planning to deliver one month after Clouds 960.  

940 – Dave D. is waiting for feedback from Norman.  Meeting planned for next Monday to 

discuss this delivery. 

974 – First round of Edition1 SSF is needed for this delivery.  May be able to deliver by the end 

of the year.  

975 – No updates. 

999 – Will be released after JIRA ticket 129 is approved. 

1001 – On track for March 14
th

 delivery.  Needed for processing Terra data from June 2005 

forward and Aqua data from March 2005 forward. 

        

Action Items 

 

Table 1. Open Action Items 

Item 

Number 

Action Item Date Status 

2014 -01 Definition and implementation of 

Configuration Codes 

02/05/2014 Lisa will work with Pam M. to determine the 

configuration code information to post on the 

Web. 

02/19/2014 Pam M. has provided feedback.  Lisa will 

incorporate Pam’s comments and then provide 

the document to Jonathan, Norman, and Dave 

D. to review. 

 

 

 

 

 

Meeting ended at 1:58 p. m. 


